Author's response to reviews

Title: Evaluation of antimicrobial, antimitagenic and radical scavenger activities of polar extracts from (Tunisian) Acacia salicina "Lindl." leaves

Authors:

Hedi Ben Mansour (hedi.mansour@hotmail.fr)

Version: 4 Date: 15 December 2011

Author’s response to reviews: see over
Editor in chief of BMC Complementary and Alternative Medicine

Dear Editor,

I have the pleasure to resubmit the revised manuscript for publication in your journal: BMC Complementary and Alternative Medicine, the revised version of our paper entitled: « Evaluation of antimicrobial, antimutagenic and radical scavenger activities of polar extracts from (Tunisian) *Acacia salicina* (Lindl) leaves» by: Jihed Boubaker, Hedi Ben Mansour, Kamel Ghedira, Leila Chekir Ghedira. Referenced: ID 3930931815928668
**Responses to Reviewer** Dario Kremer

Minor Essential Revisions:

**Title:**

Page 1: I changed ‘(Lindl.)’ into ‘Lindl.’

Page 2, line2: I changed ‘(Lindl.)’ into ‘Lindl.’

- We tried to address the grammatical and typographical errors as asked by the reviewer.

**Reviewer:** Eduardo Madrigal-Bujaidar

- We corrected the English language.